Mauna Loa Sol ar Observatory Cbserver’'s Log
Mon Cct 31 16:57:49 GV 1994
Year: 94 Doy: 304
Qobserver: yasukawa

WEATHER COMMENT: Mon Cct 31 16:58: 05 GMI 1994
Cool, in thin cirrus, light south breeze.
Tenp=42F.

LONML COWMENT: Mon Cct 31 16:59: 10 GMVI 1994

LowL is crashed. Apparently crashed at 00:37:56. C assic graphic
characters across the screen. 1’|l reload and reboot |ater after
get the rest of the systens started.

Mon Oct 31 17:04:37 GMI 1994: Patrol Start

LOML COMMENT: Mon Cct 31 17:14:15 GVIT 1994

Renoved L00125 and L00126 fromdrives 0 and 1 respectively.

Lower # tape was in drive 0, not popped out, but screen indicates
crash occurred while witing to unit 1, also not popped out.
Sonehow unit O should have ejected tape but didn't when LowL
switched over to unit 1.

Loaded L00127 and L00128 into units O and 1 respectively after

cl eani ng heads of both drives.

Rebooti ng Low L.

LONML COVMENT: Mon Cct 31 17:37:13 GMVI 1994

Low L was starting up funny. | guess | wasn't used to the new
speed in booting up, where the drives are still reloading
when the program starts. | decided to reload L0O0126 into the

drives as it was nost likely blank. Yesterday's logs indicate
that Unit O was running, crash occurred yesterday.

L00126 in unit O and LO0127 in unit 1.

Rebooti ng agai n.

COMMENT: Mon Cct 31 17:47:44 GMI 1994

STATUS: still in thin cirrus overcast, DPMON running,
LOML running snoothely, data going onto L00126 in unit O,
MKI I'l standing by.

Mon Oct 31 18:02:02 GVII 1994: Calibration
Mon Cct 31 19:01:57 GMI 1994: Calibration



MKI Tl COMMENT: Mon Cct 31 19:49:34 GMI 1994
No data on MKIII yet, still in cirrus.
Ej ected tape and did Novenmber 1 head cl eaning. Reloaded tape.

Mon Cct 31 20:01:57 GMI 1994: Calibration

WEATHER COMMENT: Mon Cct 31 20: 37:58 GMIN 1994
Crrus is thinning, orographic clouds are formng around station

MKI 11 COMMENT: Mon Cct 31 20:39:30 GMI 1994
Starting up MKIIl exavbyte.

WEATHER COMMENT: Mbn Cct 31 20:59:42 GMI 1994
I n orographic clouds.

Mon Oct 31 21:02: 03 GMI 1994: Calibration

WEATHER COMMENT: Mon Cct 31 21:19:43 GV 1994
I n orographic clouds. Stopping obs.
End

LOWL COMVENT: Mon Cct 31 21:20:30 GVI 1994
Low L crashed at 2116.

LOWL COMVENT: Mdn COct 31 21:25:42 GMVI 1994
Rermoved L00126 fromunit O, rel abel ed LO0127 fromunit 1 to O and
| oaded it into unit O, |oaded L00128 into unit 1. Rebooted.

MKI Il COMMENT: Mon Cct 31 21:31: 00 GMI 1994
Hol es between clouds are free of thin cirrus. Starting up MII
totry to get a good scan.

DPMON COMMENT: Mon Cct 31 21:31:49 GMIN 1994

Starting DPMON back up while attenpting to observe with MIII
w

Cct 31 21:32:20 GMI 1994: Patrol Start

WEATHER COMMENT: Mbn Cct 31 21:34:30 GMI 1994
UGH, the clouds are driving ne nad. | stopped DPMON and cl ouds
opened up a bit.



WEATHER COMMENT: Mon Cct 31 21:41:22 GV 1994
| think it is finally clouded up heavily enough to rel ax.

Stopping MKIII. DPMON is stopped.

COMMENT: Mon Cct 31 22:25:16 GMI 1994
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Koon taking over for Yasukawa
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WEATHER COMMENT: Tue Nov
Cl ouds never conpletely cleared,

COVNVENT: Tue Nov

1 04:15:55 GV 1994

Sent FAX of NMAP to Kevi n Reardon

Activity report:
QP: 52; 82; 105; 137; 257; 267; 274; 312.

No coronal activity.

TAPES:

MKIT'1: HO0877
DPMON:  PO0197

LONML: LO0127 in drive #0

Tue Nov 1 04:19:51 GMI 1994: Fil emark

SCAN- LOG

SCAN- LOG 17:08:06. 10/31/94 DOY 304

20:41: 38
20:57: 21
21:13: 05
21: 37: 57
0 ERRORS
K

20: 44: 48 20: 47: 55
21: 00: 29 21:03: 42
21:16: 15 21:19: 23
21:41: 06

1 04:15:15 GVI 1994
shutting down.

20: 51: 05
21: 06: 49
21: 31: 40

20:54: 12
21:09: 58
21: 34: 49



